Author’s response to reviews

Title: Giant organ confined prostatic adenocarcinoma: case report

Authors:

Jamin V Brahmbhatt (jaminbrahmbhatt@gmail.com)
Louis S Liou (liougu@yahoo.com)

Version: 4 Date: 7 December 2007

Author's response to reviews:

As requested we have performed the following formatting changes:

Formatting changes requested

-------------------------------------

Title - Please use sentence case in the title, both in the manuscript file and the submission system i.e. remove all unnecessary capitals (and no full stops/periods). For example you would change: The Title of My Prestigious Publication: A Scientific Study. to: The title of my prestigious publication: a scientific study

Title - Please ensure the manuscript title is identical in both the manuscript and submission system.

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

Qualifications - Please remove the authors’ qualifications/titles (e.g. PhD / MD / DR / Head of xx) from the manuscript and submissions system.

Affiliations - Each affiliation should be written in full and be provided in the following format: department, institute, city and country. This may mean repeating institutional addresses if the authors are from different departments within an institution. Please link each author to his or her corresponding affiliation with the use of superscript numbers.

For example:

Enitan Sawyerr1,2, Emma Veitch3
1 Plant Physiology, Zentrum für Molekularbiologie der Pflanzen (ZMBP), Universität Tübingen, Auf der Morgenstelle1, D-72076 Tübingen, Germany
2 Current address: Protein Interaction laboratory, Dept. Biochemistry and Molecular Biology, University of Southern Denmark, Campusvej 55, 5230 Odense, Denmark
3 Plant Biology, University of Minnesota, 220 Biological Sciences Center, 1445
Please remove the duplicate manuscript title on page 3. "Case Presentation: Giant Prostatic Adenocarcinoma."

Please use sentence case for all headings, for example change Case Presentation to Case presentation (lower case p) and Figure Legends to Figure legends (lower case l), etc.

Figure legends. Please reposition the Figure legends section to after the References section.

Authors. Please rename the Authors section on page 10 to Authors' contributions.

Authors' contributions. Please reposition the authors' contributions section to between the competing interests and the acknowledgements section. Please indicate that all authors read and approved the final manuscript.

References - The reference list should contain all authors' names, if there are 30 or below, the term `et al.` should not be used. If there are more than 30 names a minimum of 30 should be listed, and then et al can be used.